EO-1 Weekly Status Week of November 22 – November 28, 2007

Day of Year 326 - 332
Mission Day 2565 - 2571
Earth Observing-One (EO-1) General

There were 109 Data Collection Events (DCEs) scheduled this week.  
Two lunar calibrations were successfully executed. 

· Nominal Lunar calibration was executed during the 23:18 GMT eclipse on November 24, 2007.

· Modified Lunar was calibration executed during the 10:48 GMT eclipse on November 25, 2007.

· Preliminary analysis of the data indicates proper execution of the command load.
ALI deicing performed from 00:45 - 15:45 GMT on November 20 (day 324).

HSI out-gassing performed from 00:35 - 15:35 GMT on November 20 (day 324).

HSI instrument performed a Solar Characterization test on November 19 (day 323) at 01:05 GMT.

EO-1 Spacecraft Subsystems

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  

Instruments

All instruments operated nominally this week.  
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

Nothing new to report.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.

EO-1 MISSION OPERATIONS CENTER
Real-Time
No change from last report.

Mission Planning
· New automation concept was developed to deliver ASPEN products to the MOC.

· JPL currently delivers all products to the MOC systems via e-mail.

· Currently the SGM (Sensor Goal Monitor) software manages deliveries.   

· Periodically certain files cannot be delivered due to e-mail problems.

· New automation concept will be engineered to handle delivery, notification and processing of products.

· A combination of the secure copy function and Perl scripting will be utilized.

Flight Dynamics
No change from last report.

Data Analysis and Trending
· Performed preliminary work to study feasibility of implementing ITPS software to perform data analysis and trending in place of non-security complaint DTAS.

· Meeting will be held to discuss options and implementation on November 29, 2007 at 2:30pm in the MOC.

System Administration
No change from last report.

GROUND AND SPACE NETWORK

Station Downtimes
· No major problems to report.
Operational Discrepancies
· No operational discrepancies to report.

Support Enhancements

No change from last report.

Upcoming Events
· Orbit inclination change burns to be conducted near mid-January 2008.

Imagery Status
Scenes and Engineering Cals planned for week of Nov 22 – Nov 28, 2007                          109
Total scenes and engineering calibrations planned for entire mission – approximately 47,257
Total Scenes:  ALI scenes in the Level 0 archive            33,717 (as of Nov 28, 2007)

                        Hyperion scenes in the Level 0 archive    33,467
Publications and Presentations Status (as of 07-05-07)
404 publications 

284 external presentations

51 articles and press releases
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